Author’s response to reviews

Title: Presentation of a keratocystic odontogenic tumor with agenesis: a case report.

Authors:

Mariano ML Lacarbonara (mariano.lacarbonara@libero.it)
Giuseppe GM Marzo (giuseppe.marzo@cc.univaq.it)
Vitantonio VL Lacarbonara (antlac@tiscali.it)
Annalisa AM Monaco (annalisamonaco@yahoo.it)
Mario MC Capogreco (mario.capogreco@cc.univaq.it)

Version: 6 Date: 28 January 2014

Author’s response to reviews: see over
Dear Editor-in-Chief,

I’m sending you our revised manuscript entitled “ODONTOGENIC KERATOCYST WITH AGENESIS: A CASE REPORT.” by Lacarbonara Mariano, Marzo Giuseppe, Lacarbonara Vitantonio, Monaco Annalisa e Capogreco Mario from the University of L’Aquila, Italy.

We modified the case report as suggested by reviewers.

You can find all the changes underlined in yellow.

The references are now in Journal style and the English was revised by Write Science Right.

Please let me know your decision at the earliest of your convenience.

With my best regards

Mariano Lacarbonara